Author's response to reviews

Title: Common mental disorder in Nyanza province, Kenya - prevalence and associated risk factors in 2013 - an assessment of change since 2004, using a repeat household survey in a demographic surveillance site

Authors:

Rachel Jenkins (rachel@olan.org)
Caleb Othieno (ciothieno@uonbi.ac.ke)
Linnet Ongeri (longeri@kemri.org)
Peter Sifuna (psifuna@yahoo.com)
Michael Ongecha (michaelongecha@yahoo.com)
David Kiima (dmkiima@gmail.com)
James Kingora (jkmboroki@yahoo.com)
Raymond Omollo (romollov@yahoo.co.uk)
Bernhards Ogutu (ogutu6@gmail.com)

Version: 4
Date: 6 November 2015

Author's response to reviews:

Dear editor

thank you for the latest reviewer reports and for your editorial requirements, which we have responded to as below

1. Authors' contributions

Please ensure that all authors are explicitly mentioned. Currently, author MO is not included in this section.
DONE

2. References

Please note that reference 13 is incomplete. Please also consult the BMC Psychiatry style for citing online sources (http://www.biomedcentral.com/bmcpsychiatry/authors/instructions/researcharticle#formatting-references).
In particular, please add a statement about when each site was assessed.
DONE

3. Figure

Please state in the cover letter whether the map depicted in your figure is your own or taken from another source. If taken from another source please acknowledge the source(s) in the figure legend, and if they are under copyright also state the written permissions given to use and adapt them.

If the above conditions are not met the image needs to be removed. Please note that the editors may request proof of permissions at any time.

Should you require an alternative source you may wish to try Wikimedia
Commons:http://commons.wikimedia.org/wiki/Main_Page
THE FIGURE IS PRODUCED AND SUPPLIED BY OUR CO-AUTHOR PETER SIFUNA
4. Title page
Please ensure that the email addresses of all authors are provided on the title page. DONE